Author’s response to reviews

Title: An Australian Aboriginal birth cohort: a unique resource for a life course study of an indigenous population. A study protocol

Authors:

- Dr Susan Sayers (sue@menzies.edu.au)
- Dorothy Mackerras (dorothy@menzies.edu.au)
- Gurmeet Singh (gurmeet@menzies.edu.au)
- Ingrid Bucens (ingbucens@hotmail.com)
- Mrs Kathryn Flynn (pharlap@octa4.net.au)
- Alison Reid (alisonreid@dph.uwa.edu.au)

Version: 5 Date: 27 Feb 2003

I have just discovered another typing mistake. How do these things sneak past all the checking? On page 11 in the data collection paragraph third line from the bottom of the page. Sentence beginning The team worked predominately ..... but other buildings, verandas, and private homes and health clinics were all used.